Author’s response to reviews

Title: Necrotizing Fasciitis following measles vaccine administration: A case report

Authors:

ELVIS ISERE (elvisisere@gmail.com)
AKINOLA FATIREGUN (akinfati@yahoo.com)
OMOLAYO OLUBOSEDE (oolubosede@yahoo.com)
MODUPEOLA DOSUMU (dupedosumu@gmail.com)
OLASEINDE BELLO (laseinde@yahoo.com)

Version: 2 Date: 31 May 2019

Author’s response to reviews:

Editors comment

1. Please combine your "Discussion" and "Conclusions" sections into one section entitled "Discussion and Conclusions".

   Response-Discussion and conclusion section have been merged together

2. Please remove any additional files you do not want published alongside your manuscript, including cover/response letters.

   Response-These have been unchecked and removed

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

   Response-Done
Reviewer 1 comments

1. The abstract that appears on the cover page of the manuscript is from the original submission, not the revised version that appears in the manuscript file. Please correct.

   Response-This have been corrected

2. Causality assessment

   Page 4, line 96: "The findings from the investigation..." - this should be a new paragraph.

   Response-Done

3. page 4, lines 109-110: "...blood samples collected by the attending physician during the preliminary case management at a local hospital" - It may not be clear which patient is being referred to here, suggest clarifying that this refers to the child with NF, for example: "blood samples collected on the child with NF by the attending physician at the local hospital...."

   Response-Correction done

4. page 4, line 111: "microbiolgical" should be "microbiological"

   Response-Correction done

5. Discussion: The authors did not address Reviewer 2's comment regarding possibility of other organisms besides S. aureus (particularly Group A Streptococcus) and his request to mention the appropriate antibiotic management of NF, for example addition of clindamycin. These comments should be addressed.

   Response- The authors response have been included

Regards

Elvis Isere

Corresponding author